

Metamorphic peak estimates of the Marguareis Unit (Briançonnais Domain): New constrains for the tectonic evolution of the south-western Alps

Edoardo Sanità, Maria Rosa, Jean-Marc Lardeaux, Michele Marroni, Luca Pandolfi

▶ To cite this version:

Edoardo Sanità, Maria Rosa, Jean-Marc Lardeaux, Michele Marroni, Luca Pandolfi. Metamorphic peak estimates of the Marguareis Unit (Briançonnais Domain): New constrains for the tectonic evolution of the south-western Alps. Terra Nova, In press, 10.1111/ter.12592. hal-03636199

HAL Id: hal-03636199 https://hal.science/hal-03636199v1

Submitted on 9 Apr 2022

HAL is a multi-disciplinary open access archive for the deposit and dissemination of scientific research documents, whether they are published or not. The documents may come from teaching and research institutions in France or abroad, or from public or private research centers. L'archive ouverte pluridisciplinaire **HAL**, est destinée au dépôt et à la diffusion de documents scientifiques de niveau recherche, publiés ou non, émanant des établissements d'enseignement et de recherche français ou étrangers, des laboratoires publics ou privés.

RESEARCH ARTICLE

Terra Nova Wiley

Metamorphic peak estimates of the Marguareis Unit (Briançonnais Domain): New constrains for the tectonic evolution of the south-western Alps

Edoardo Sanità^{1,2} | Maria Di Rosa² | Jean Marc Lardeaux^{3,4} | Michele Marroni^{2,5} | Luca Pandolfi^{2,5}

¹Dipartimento di Scienze della Terra, Università di Firenze, Firenze, Italy

²Dipartimento di Scienze della Terra, Università di Pisa, Italy

³Géosciences Azur, UMR CNRS Université de Nice-Sophia Antipolis, Nice, France ⁴Centre for Lithospheric Research, Czech Geological Survey, Prague 1, Czech

Geological Survey, Prague 1, Czech Republic

⁵Consiglio Nazionale della Ricerca, Istituto di Geoscienze e Georisorse, 1 Pisa, Italy

Correspondence

Maria Di Rosa, Università di Pisa, Dipartimento di Scienze della Terra, Via Santa Maria, 53, Pisa, Italy. Email: maria.dirosa.scaglia@gmail.com

Funding information PRA project - Università di Pisa

Abstract

In the Western Alps, while many *P-T* constraints are available for the high-grade units, the dataset for the low-grade units located along the westernmost edge of the belt remains scarcer and incomplete. To face these uncertainties, we quantify for the first time the metamorphic peak conditions of the Marguareis Unit, a fragment of European margin (Briançonnais Domain) regarded till now as very-low-grade unit, exposed along the boundary between Maritime and Ligurian Alps (South-Western Alps), whose deformation history is well constrained. The Marguareis Unit is regarded as a thinned portion of the European continental margin involved in the Alpine collision. Metamorphic peak conditions reconstructed using the chlorite-phengite multiequilibrium approach on the Eocene siliciclastic deposits indicate that the Marguareis Unit recorded high-pressure and low-temperature conditions during its subduction, accretion and subsequent exhumation into the Alpine wedge.

1 | INTRODUCTION

It was widely recognized that quantitative pressure-temperaturetime (P-T-t) paths extracted from metamorphic units portray the thermo-mechanical evolutions of orogens (e.g. England and England & Thompson, 1984; England & Richardson, 1977). The latter are usually reconstructed by means of the P-T conditions of highgrade units. However, the reconstruction of P-T paths, especially high-pressure metamorphism, in low- to very low-grade metamorphic rocks, is of paramount importance to better understand the geodynamic evolution of orogens. Indeed, under low-temperature condition reaction, kinetics is sluggish and consequently chemical equilibrium is rarely achieved at the whole-rock scale impeding to unambiguously constrain P-T conditions of metamorphism (Frey & Robinson, 1999; Lanari & Engi, 2017). It is all the more difficult to determine *P*-*T* conditions in meta-sediments in which detrital minerals remain as relict phases, so making the so-called reactive rock composition (Lanari & Engi, 2017 and reference therein), to be taken into account in order to model the chemical equilibrium, still questionable.

This is the case of the very low-grade Marguareis Unit (Briançonnais Domain), a European fragment of continental crust (Decarlis et al., 2013), located at the boundary between Maritime and Ligurian Alps. This unit was involved in the Alpine collision since middle Eocene age (Seno et al., 2005). The deformation history of the Marguareis Unit has been deeply investigated (Brizio et al., 1983; Sanità et al., 2021), whereas the reconstruction of metamorphic conditions remains partly unconstrained and interpreted as the result of very shallow tectonic events (anchizone, Messiga et al., 1981). Contrarily, some authors (i.e. Lanari et al., 2014; Michard et al., 2004;

This is an open access article under the terms of the Creative Commons Attribution License, which permits use, distribution and reproduction in any medium, provided the original work is properly cited.

© 2022 The Authors. Terra Nova published by John Wiley & Sons Ltd.

² WILEY- Terra

Strzerzynski et al., 2011) reconstructed high pressure-low temperature (HP-LT) metamorphic paths for the Briançonnais Units cropping out in other sectors of the Alpine belt. The Marguareis Unit being located at the top of a SW-verging stack of low- to very lowgrade units (Brizio et al., 1983; Sanità et al., 2020). Particularly, above the Late Cretaceous non-metamorphic Helminthoid Flysch Unit (Figures 1 and 2), the quantification of metamorphic peak conditions is crucial to reconstruct the vertical motions of tectonic units during the edification of the Alpine wedge of the South-Western Alps.

Our goal is to propose the first quantification of the peak conditions, and subsequent P-T evolution, for the Marguareis Unit obtained after detailed meso- to microstructural investigations. This involved a careful identification of sites where local chemical equilibrium was possibly achieved. In these, the composition of chlorite and phengite pairs was assumed to be in equilibrium; here, use for the P-T determination using the chlorite-phengite (Chl, Phg) multiequilibrium approach (Lanari, 2012; Vidal et al., 2006) is a powerful tool for P and T calculations in meta-sediments (minerals abbreviations are from Warr, 2021).

1.1 Geological context and deformation history

The Marguareis Unit (Monte Marguareis area-Figures 1 and 2) is tectonically stacked with units stemming from Ligure Piemontese Ocean and the neighbouring Europa continental margin (Vanossi, 1986; Vanossi et al., 1984). Its finite structural pattern results from a deformation history mainly dealing with superposed fold systems developed at different structural levels (Sanità et al., 2020, 2021). However, some authors regarded the structural architecture of the area as the result of a faulting event that is part of an E-W-direct shearing (Bertok et al., 2018; D'Atri et al., 2016; Piana et al., 2009). According to Sanità et al. (2021), this unit recorded two micro- to map-scale folding events (D1 and D2, Figure 3), and subsequently, mostly developed at large scale, D3 and D4 folding systems (Figures 2 and 3, syn-stacking and post-stacking structures—Sanità et al., 2020), regarded as the results of underthrusting and accretion into the Alpine collisional wedge (Brizio et al., 1983; Carminati, 2001; Carminati & Gosso, 2000; Sanità et al., 2020, 2021). The D1 phase produced similar and isoclinal SW-verging folds with scattered fold axes and NW-SE striking axial planes, whereas the D2 phase developed similar-to-parallel tight NE-verging folds with NW-SE trending fold axes and NW-SE striking axial planes. During the D1 and D2 phases, metamorphic re-crystallization took place along the S1 and S2 foliations associated with the F1 and F2 folding systems (Figure 3). S1 is a slaty cleavage marked by the preferred orientation of $Chl + Phg + Qz + Ab + K - Fsp + Ep \pm Cal metamorphic assem$ blage and stylolitic surfaces (Figure 3). Within the S1, chlorite grains are stocky, sometimes with re-crystallization tails, and reach 30 µm in size, while phengites are more elongated and range in length between 20 and 30µm. Both shows undulose extinction and sharp edges. The S2 is a crenulation cleavage (Figure 3) marked by synmetamorphic Phg+Chl+Qz+Ab+K-Fsp±Cal association. Both Chl

Significance statement

We quantify the P-T conditions recorded by the Europeanderived Marguareis Unit (Brianconnais Domain) cropping out along the south-western sector of the Marguareis Massif (South-Western Alps). This unit was buried at depth corresponding to blueschists facies conditions during the Alpine collision. The reconstruction of the P-T path of the Marguareis Unit permitted to better constrain the tectonic evolution of this sector of the South-Western Alps.

and Phg along the S2 are thin, not exceeding 10 μ m in length and show sharp edges. Big altered crystals of detrital chlorite (~80µm in length) and white mica (40µm in length) both characterized by frayed edges, chemical zoning and re-crystallization tails are also present (Figure 3, Material S1).

The subsequent D3 and D4 folding events produced a regionalscale NE-SW trending fold with north-eastward dipping axial plane (Sanità et al., 2020) and open-fold system with sub-horizontal axial planes (Figure 3) respectively. The associated tectonic foliations marked by stylolitic surfaces without evidence of metamorphic recrystallization suggest their development at shallower structural levels (Sanità et al., 2021).



FIGURE 1 (a) Geographic location of the study area. (b) Simplified geological sketch of the South-Western Alps (modified and redraw by Molli et al., 2010) with regional-scale geological cross-section (redraw and modified from Bonini et al., 2010). Black box indicates the study area of Figure 2. In the cross-section, red line indicates the Pennidic front. TPB, Tertiary Piedmont Basin; Oph, Alpine Ophiolitic Units; MT, Moglio-Testico Unit; DM, Dora Maira Massif; Ar, Argentera Massif



FIGURE 2 Structural sketch of the south-western sector of the Marguareis Massif with geological cross-section. Stereographic plots show the spatial relationships between planar and linear structural features recognized in the Marguareis Unit. The stratigraphic log (Middle Triassic-Middle Eocene age) of the Marguareis Unit is reported on the right corner (by Sanità et al., 2020). Red and yellow stars indicate the sample location

The last deformation event is represented by high angles normal to transtensional faulting that did not produce profound modification in the previous architecture.

2 | METHODS

Meso-structural analysis combined with rigorous microstructural observations aimed to attest for costability of a given mineral pair, in our case chlorite and white mica, which allowed us to reasonably assume that conditions of local chemical equilibrium were attained and calculate P and T conditions of this equilibrium. Thus, calibrating X-ray maps on two samples collected from the middle Eocene metapelites of the Marguareis Unit (ED1 and TSE236), we closely linked the composition of chlorite and phengite to microstructural domains (S1 and S2). For each chlorite-phengite pairs grown along the S1 an S2 foliation planes, we calculate the temperature and pressure estimations by means of XMapTools software (Lanari et al., 2014) and ChIMicaEqui software (Lanari, 2012, the description of the method is in the Material S1). Our results are compared to classical chlorite thermometry (the chemical analyses of chlorite and phengite are provided in the Material S1). This has been progressively improved (Bourdelle & Cathelineau, 2015; Lanari et al., 2014) following the pioneering calibrations of Cathelineau and Nieva (1985) and Cathelineau (1988) and successfully applied in low-grade

metamorphic and/or hydrothermal systems (e.g. Beauchamps et al., 2021; Bourdelle, 2021; Favier et al., 2021; Vérati et al., 2018).

3 | RESULTS

Combining microstructural evidence with the compositional variability of chlorite (Figures S1-S3), three different compositions have been distinguished. The first composition is related to detrital chlorites which remain metastable during Alpine metamorphism. The second and the third compositions are related to chlorite re-crystallized along the S1 and S2 foliations respectively. D1 chlorites (Chl1) are characterized by low Si content (2.65-2.80 a.p.f.u.), and Al content which vary from 2.60 to 3 a.p.f.u. in both samples (Figure 4a and Material S1). D2 chlorites (Chl2) are aligned on the S2 and have Al content which is basically low (2.40-2.90 a.p.f.u.), whereas Si content is higher (2.75-3.00 a.p.f.u.) than those of Chl1 (Figure 4a and Material S1). The differences between the two samples are more evident by observing the composition of end-members: clinochlore+daphnite content is always higher than 50% in ED1, while in TSE236 it is lower than 40% in the Chl1 and range between 50% and 40% in the Chl2. In the sample ED1, small differences between Chl1 and Chl2 are given by the Amesite content which is higher and lower than 20% respectively. In both samples, XMg in chlorite tends to be higher in Chl1 than those of Chl2, even if the total range is between 0.37 and 0.45.



FIGURE 3 Meso- and micro-scale D1- and D2-related structural features of the Marguareis Unit. F1 and F2 folds (a, b) in the middle Eocene meta-pelites. (c-f) Microphotographs of the S1 and S2 foliations. The cross-cutting relationships are reported. AP1: F1-related axial plane; AP2: F2-related axial plane; AP4: F4related axial plane

Terra Nova_-Wiley

Similarly, phengite composition varies according to the different microstructural sites (S1 and S2 foliations and detrital phengites, Figure 3c, Figures S1–S3). Al and Si contents in mica related to the D1 (Phg1) and D2 (Phg2) phases are quite homogeneous into the two samples (Si: 2.90–3.40 a.p.f.u. in ED1, 2.90–3.20 a.p.f.u. in TSE236; Al: 2.35–2.60 a.p.f.u. in ED1, 2.30–2.70 a.p.f.u. in TSE236) (Figure 4b). In both the samples, a clear trend from Si-rich–Na-poor Phg1 to Si-poor–Na-rich Phg2 is identified. Phengites of ED1 sample are characterized by celadonite content always lower than 20% with

FIGURE 4 Geochemistry of (a) chlorite and (b) phengite sampled into ED1 and TSE236. Chlorite and phengite structural formula are calculated assuming 14 and 11 oxygen, respectively. (c) P/T diagram showing the results of the thermobaric methods applied to samples ED1 and TSE236. In the P/T space, the pale squares indicate the areas constrained by the Chl-Qz-wt method (Vidal et al., 2006) and the Phg-Qz-wt method (Dubacq et al., 2010); the position of data indicates the P-T equilibrium conditions of single Chl-Phg couples obtained with Chl-Phg-Qz-wt method (Vidal & Parra, 2000). Histograms along the Y-axis indicate the results of Massonne and Schreyer (1987) geobarometer. Along the X-axis, the used thermometers are reported with range values related to D1 and D2 phases

Phg1, which tend to be more muscovitic. In the sample TSE236, the celadonite content is more variable (higher in Phg1 than Phg2) but never exceeds 35%. However, both Phg1 and Phg2 are classifiable as impure muscovite.

P-*T* estimates based on the activity of chlorite and phengite endmembers and the water activity were obtained through the Chl-Qz-wt, Phg-Qz-wt and Chl-Phg-Qz-wt methods. These methods were applied on chlorites and phengites related to the D1 phase in the sample ED1 and those of the D2 phase in the sample TSE236.





FIGURE 5 (a) P-T path of the Marguareis Unit (black arrow in the P-T space) obtained in this study. The paths related to the HP Brianconnais (grey arrow) are also plotted (data from Michard et al., 2004). The microdomains, in which the Chl-Phg couples used to constrain the P-T conditions of D1-D2 phases were sampled, are evidenced (b) in the sample ED1 with the green circle and (c) in the sample TSE236 with black rectangle. In (b) and (c), chlorite shadows are referred to the Fe-content (green-yellow-red colours) and phengite shadows represent the K-intensity (blue to white colours); other mineral phases (quartz, calcite, titanite and feldspar) are in grey

With the Chl-Qz-wt (Vidal et al., 2006), T ranges were calculated setting the equilibrium tolerance to 30°C and the percentage of Fe³⁺ for each Chl analysis fixing a starting pressure value (in this case, 0.6 GPa) following the procedure of Lanari and Duesterhoeft (2019). The water activity was set to 1 in the sample ED1 and to 0.8 in the sample TSE236 based on the presence/lack of calcite (Frassi et al., 2022). Temperature ranges related to the D1 and D2 phases are 230-330°C and 160-330°C respectively. The Phg-Qz-wt (Dubacg et al., 2010) was applied in order to quantify the P conditions of the Marguareis Unit during the D1 and the D2 phases. The P range related to the D1 phase was estimated with Phg1 of the sample ED1 grown along the S1, setting an average temperature of 300°C and assuming 0.95% of water as optimized value (Material S1, Frassi et al., 2022). Pressure range for the D1 phase is between 1.05 and 0.65 GPa (Figure 4c). Pressure values related to the D2 phase were calculated using Phg2 of the sample TSE236, assuming an average T of 300°C and the water optimized values of 0.95% (Material S1, see Frassi et al., 2022). Pressure range of the D2 phase is 0.38-1.10 GPa (Figure 4c).

The Chl-Phg-Qz-wt (Vidal & Parra, 2000) method allowed to find the chlorite-phengite couples related to the D1 (sample ED1) and D2 (sample TSE236) phases (see Material S1). Only the couples equilibrated with a tolerance <1000 J and included between the *T*-and *P*-ranges of the Chl-Qz-wt and Phg-Qz-wt method were considered. The *P*-*T* conditions for the D1 phase are 1.00-0.90 GPa and 280-330°C and those related to the D2 phase are 0.85-0.7 GPa and 230-300°C (Figure 4c).

The results obtained were compared with T and P values estimated with classical thermobarometry (Figure 4c). We first used the semi-empirical calibration proposed by Cathelineau (1988) and this thermometer yields a temperature range 275-327°C for chlorites in S1 and 215-300°C for chlorites in S2. However, Lanari et al. (2014) proposed a calibration for low-temperature conditions (i.e. T < 400°C) even when the amount of Fe³⁺ in chlorite is unknown. Using this calibration, we obtained a temperature range 310-330°C for chlorites in S1 and a *T* range 250-280°C for chlorites in S2. Moreover, we used also the calibration established by Bourdelle and Cathelineau (2015), which is also an efficient tool for chlorite thermometry between 100 and 400°C, even if the non-ideal contribution of site mixing is not considered. This thermometer yields temperatures of 275-285°C for chlorite in S1 and of 225-275°C for chlorites in S2. Altogether, chlorite solid-solution thermometry indicates *T* conditions of $300\pm30°$ C for the D1 phase and *T* conditions of $250\pm40°$ C for the D2 phase.

The geobarometer of Massonne and Schreyer (1987) was applied on the samples ED1 and TSE236 (XMapTools, Lanari et al., 2014) to obtain the pressure range related to the D1 and D2 phases respectively (Figure 4c). The pressure range calculated on phengite of the sample ED1 grown along the S1 is 0.63–1.03 GPa (>3000 pixel/map), while that of the sample TSE236 related to the S2 is 0.38–1.07 GPa (>1000 pixel/map).

4 | DISCUSSION

The polyphase deformation history recorded in the Marguareis Unit is regarded as the result of underthrusting, accretion and subsequently exhumation into the Alpine wedge (Brizio et al., 1983; Carminati, 2001; Sanità et al., 2021). This structural pattern is marked by the superposition of folding systems (D1-D4) developed at different structural levels. Only during the D1 and D2 phases, metamorphic re-crystallization chlorite and phengite grains occurred. The thermobarometric estimations performed in this study show that the prograde path is not recorded and the chlorite-phengite pairs grown along S1 and S2 foliations depict the retrograde path of the Marguareis Unit (Figure 5). The exhumation already started during the D1 phase and it continues during the D2 phase up to shallower crustal levels into the Alpine wedge. P-peak conditions of 1.00-0.90 GPa and 280-330°C were registered by phengite-chlorite couples related to D1 phase (Figures 4c and 5a). The retrograde path of the Marguareis Unit is constrained by the re-crystallization of chlorite and phengite during the D2 phase at 0.85-0.7 GPa and 230-300°C (Figure 5a). No significative heating events during the D2 exhumation phases were identified from the thermobarometry for the Marguareis Unit. The different p values constrained on Phg1 and Phg2 are considered as the witnesses of the vertical movement of the Marguareis Unit inside the Alpine wedge. Thus, the reconstructed pathway strongly suggested that the metamorphic imprint of the Marguareis Unit was recorded in a continental subduction setting, the latter already documented in other similar sectors of the Alpine belt (Di Rosa et al., 2019; Lanari et al., 2012; Michard et al., 2004; Sterzynsky et al., 2011). Our results allow us to re-interpret the Margureis Unit, till now regarded as a very low-grade unit (Messiga et al., 1981), as a HP-LT unit. In the South-Western Alps, HP-LT units were documented by others authors (Goffé et al., 2004; Lardeaux, 2014), and similar HP-LT paths were reconstructed for the Brianconnais Units in more internal sector of the belt (Lanari et al., 2014; Sterzynsky et al., 2011).

Michard et al. (2004) reconstructed the P-T path for the Brianconnais Units outcropping in the Embrunais-Ubaye area. The authors proposed that the Brianconnais Units recorded their metamorphic peak during the D1 phase (HP Brianconnais path in Figure 5a) immediately followed by the onset of the exhumation which begins at the end of D1 phase. The P-T path proposed for the HP Brianconnais Units by Michard et al. (2004) shows clear similarities with those we propose for the Marguareis Unit (Figure 5a). For the HP Briançonnais Units, the late Bartonian age is proposed (Monié, 1990) as age constraint for the metamorphic peak conditions (similar timing was proposed by Lanari et al., 2012). Therefore, based on the close relationships between Marguareis Unit and the Brianconnais Units in this work, we can suppose the Bartonian age as timing for the achievement of the metamorphic peak conditions also for the Marguareis Unit (as suggested by Sanità et al., 2020). The final exhumation of the Marguareis Unit through the D3 and D4 folding events as well as the faulting concludes the geological framework from Late Eocene onward.

5 CONCLUSIONS

We quantified the metamorphic peak conditions and P-T evolution of the Marguareis Unit (Briançonnais Domain) exposed at the Nova_-WILEY

boundary between Maritime and Ligurian Alps (South-Western Alps). Our results indicate that:

- The reconstructed P-T path suggests that the Marguareis Unit experienced HP-LT metamorphism during the building of the Alpine belt (P-peak: 1.00-0.90 GPa and T-peak: 280-330°C), and should therefore no longer be considered as the only Brianconnais Unit affected by anchizone metamorphism.
- The Marguareis Unit preserved only evidence of retrograde (decompression) metamorphism into the Alpine wedge.
- The metamorphic peak conditions were achieved during the D1 deformation stage. The exhumation started during the late D1 and continued during the D2 phase as already proposed by previous authors for Brianconnais Units exposed in other sectors of the belt. The exhumation towards shallower structural levels concludes with the subsequent D3 and D4 folding and faults.

ACKNOWLEDGEMENTS

A special thanks to Guido Gosso to have stimulated this kind of work as well as critical suggestions to perform it. We thank Andrea Risplendente for the technical support during EMPA data acquisition. We thank Pierre Lanari and an anonymous reviewer and the Editor Carlo Doglioni for their constructive revisions. This study was supported by University of Pisa (PRA project). Open Access Funding provided by Universita degli Studi di Pisa within the CRUI-CARE Agreement.

DATA AVAILABILITY STATEMENT

The data that support the findings of this study are available from the corresponding author upon reasonable request.

ORCID

Edoardo Sanità D https://orcid.org/0000-0002-5488-9832 Maria Di Rosa (D) https://orcid.org/0000-0002-1154-7429 Jean Marc Lardeaux D https://orcid.org/0000-0001-7666-7109 Michele Marroni D https://orcid.org/0000-0002-2947-3739 Luca Pandolfi D https://orcid.org/0000-0002-6129-647X

REFERENCES

- Beauchamps, G., Bourdelle, F., Dubois, M., Hebert, R. L., & Ledésert, B. A. (2021). First characterization of the cooling of the paleogeothermal system of Terre-de-haut (les Saintes archipelago, Guadeloupe): Application of fluid inclusion and chlorite thermometry. Journal of Volcanology and Geothermal Research, 419, 107370.
- Bertok, C., Musso, A., D'Atri, A., Martire, L., & Piana, F. (2018). Geology of Colle di Tenda - Monte Marguareis area (Ligurian Alps NW Italy). Swiss Journal of Geosciences, 104(2), 237-255.
- Bonini, L., Dallagiovanna, G., & Seno, S. (2010). The role of pre-existing faults in the structural evolution of thrust systems: Insights from the Ligurian Alps (Italy). Tectonophysics, 480, 73-87.
- Bourdelle, F., & Cathelineau, M. (2015). Low-temperature chlorite geothermometry: A graphical representation based on a T-R2+-Si diagram. European Journal of Mineralogy, 27, 617–626.
- Bourdelle, F. (2021). Low-temperature chlorite geothermometry and related recent analytical advances: A review. Minerals, 2021, 130.

- Brizio, F. D., Deregibus, C., Eusebio, A., Gallo, M., Gosso, G., Rattalino, E., & Oxilia, M. (1983). Guida all'escursione: i rapporti tra la zona Brianzonese Ligure e il Flysch a Elmintoidi, Massiccio del Marguareis. (Limone Piemonte-Certosa di Pesio, 468 Cuneo, 14/15 Settembre 1983). Memorie della Società Geologica Italiana, 26, 579-595.
- Carminati, E., & Gosso, G. (2000). Structural map of a Ligurian Briançonnais cover nappe (Conca delle Carsene, Monte Marguareis, Ligurian Alps, Italy) and explanatory notes. *Mem. Soc. Geol.*, 52, 93–99.
- Carminati, E. (2001). Incremental strain analysis using two generations of syntectonic coaxial fibres: An example from the Monte Marguareis Briançonnais cover nappe (Ligurian Alps, Italy). *Journal of Structural Geology*, 23, 1441–1456.
- Cathelineau, M., & Nieva, D. (1985). A chlorite solid solution geothermometer the los Azufres (Mexico) geothermal system. *Contributions* to Mineralogy and Petrology, 91, 235–244.
- Cathelineau, M. (1988). Cation site occupancy in chlorites and illites as a function of temperature. *Clay Minerals*, *23*, 471–485.
- D'Atri, A., Piana, F., Barale, L., Bertok, C., & Martire, L. (2016). Geological setting of the southern termination of Western Alps. International Journal of Earth Sciences, 105, 1831–1858.
- Decarlis, A., Dallagiovanna, G., Lualdi, A., Maino, M., & Seno, S. (2013). Stratigraphic evolution in the Ligurian Alps between Variscan heritages and the alpine Tethys opening: A review. *Earth-Science Reviews*, 125, 43–68.
- Di Rosa, M., Frassi, C., Meneghini, F., Marroni, M., Pandolfi, L., & De Giorgi, A. (2019). Tectono-metamorphic evolution in the European continental margin involved in the alpine subduction: New insights from the alpine Corsica, France. *Comptes Rendus Geoscience*, 351, 384–394.
- Dubacq, B., Vidal, O., & De Andrade, V. (2010). Dehydration of dioctahedral aluminous phyllosilicates: Thermodynamic modelling and implications for thermobarometric estimates. *Contributions to Mineralogy and Petrology*, 159, 159–174.
- England, P. C., & Richardson, S. W. (1977). The influence of erosion upon the mineral facies of rocks from different metamorphic environments. *Journal of the Geological Society of London*, 134, 201–213.
- England, P. C., & Thompson, A. B. (1984). Pressure-temperature-time paths of regional metamorphism: Heat transfer during the evolution of regions of thickened continental crust. *Journal of Petrology*, 25, 894–928.
- Favier, A., Verati, C., Lardeaux, J. M., Münch, P., Renac, C., Corsini, M., & Orange, F. (2021). 40Ar/39Ar dating of high temperature geothermal systems: First attempt on hydrothermally altered pyroxenes from the Saintes archipelago (Lesser Antilles arc, Guadeloupe). *Chemical Geology*, 581, 120401.
- Frassi, C., Di Rosa, M., Farina, F., Pandolfi, L., & Marroni, M. (2022). Anatomy of a deformed upper crust fragment from western alpine Corsica (France): Insights into continental subduction processes. *International Geology Review*, 64, 1–21. https://doi. org/10.1080/00206814.2022.2031315
- Frey, M., & Robinson, D. (1999). Low-grade metamorphism. Blackwell Science.
- Goffé, B., Schwartz, S., Lardeaux, J. M., & Bousquet, R. (2004). Metamorphic structure of the western and Ligurian Alps. In R. Oberhansli (Ed.), *Explanatory note to the map "metamorphic structure of the Alps"* (pp. 125–144). Mitt. Österr. Mineral. Gesell.
- Lanari, P., 2012. Micro-cartgraphie P-T-dans les roches metamorphiques. Applications aux Alpes et à l' Himalaya. PhD thesis. Université de Grenoble.
- Lanari, P., Guillot, S., Schwartz, S., Vidal, O., Tricart, P., Riel, N., & Beyssac, O. (2012). Diachronous evolution of the alpine continental subduction wedge: Evidence from P-T estimates in the Briançonnais zone houilleré (France-Western Alps). Journal of Geodynamics, 56, 39–54.
- Lanari, P., Vidal, O., De Andrade, V., Dubacq, B., Lewin, E., Grosch, E., & Schwartz, S. (2014). XMAPTOOLS: A MATLAB c -based program

for electron microprobe X-ray image processing and geothermobarometry. *Computers and Geosciences*, *62*, 227–240.

- Lanari, P., & Engi, M. (2017). Local bulk composition effects on metamorphic mineral assemblages. *Reviews in Mineralogy and Geochemistry*, 83, 55–102.
- Lanari, P., & Duesterhoeft, E. (2019). Modeling metamorphic rocks using equilibrium thermodynamics and internally consistent databases: Past achievements, problems and perspectives. *Journal of Petrology*, 60, 19–56.
- Lardeaux, J. M. (2014). Deciphering orogeny: A metamorphic perspective. Examples from European alpine and Variscan belts. *Bulletin. Société Géologique de France*, 185, 93–114.
- Massonne, H. J., & Schreyer, W. (1987). Phengite geobarometry based on the limiting assemblage with K-feldspar, phlogopite, and quartz. *Contributions to Mineralogy and Petrology*, 96, 212–224.
- Messiga, B., Oxilia, M., Piccardo, G. B., & Vanossi, M. (1981). Fasi metamorfiche alpine nel Brianzonese e Prepiemontese esterno delle Alpi liguri: un possibile modello evolutivo. *Rend. SIMP, 38*, 261-280.
- Michard, A., Avigad, D., Goffé, B., & Chopin, C. (2004). The highpressure metamorphic front of the southwestern Alps (Ubaye-Maira transect, France, Italy). Schweizerische Mineralogische und Petrographische Mitteilungen, 84, 215–235.
- Molli, G., Crispini, L., Malusà, G. M., Mosca, M. G., Piana, F., & Federico, L. (2010). Geology of the Western Alps-northern Apennine junction area – A regional review. *Journal of the Virtual Explorer*, 36, 1–49.
- Monié, P. (1990). Preservation of Hercynian 40Ar/39Ar ages through high-pressure low-temperature alpine metamorphism in the western alps. *European Journal of Mineralogy*, *2*, 343–361.
- Piana, F., Musso, A., Bertok, C., D'Atri, A., Martire, L., Perotti, E., ... Martinotti, G. (2009). New data on post-Eocene tectonic evolution of the external Ligurian Briançonnais (Western Ligurian Alps). *Italian Journal of Geosciences*, 128, 353–366.
- Sanità, E., Lardeaux, J. M., Marroni, M., Gosso, G., & Pandolfi, L. (2020). Structural relationships between Helminthoid Flysch and Briançonnais units in the Marguareis massif: A key for deciphering the finite strain pattern in the external southwestern Alps. *Geological Journal*, 56, 2024–2040. https://doi.org/10.1002/gj.4040
- Sanità, E., Lardeaux, J. M., Marroni, M., Gosso, G., & Pandolfi, L. (2021). Deciphering large-scale superoposed fold systems at shallow crustal levels in collision zones: Insights from the Marguareis massif (southwestern Alps). *Journal of Maps*, 17, 559–568.
- Seno, S., Dallagiovanna, G., & Vanossi, M. (2005). A kinematic evolutionary model for the Penninic sector of the central Ligurian Alps. International Journal of Earth Sciences, 94, 114–129.
- Strzerzynski, P., Guillot, S., Leloup, P. H., Arnaud, N., Vidal, O., Ledru, P., Corrioux, G., & Darmendrail, X. (2011). Tectono-metamorphic evolution of Brianconnains zone (Modane-Aussois and southern Vanoise units, Lyon Turin transect, Western Alps). Journal of Geodynamics, 56-57, 55-75.
- Vanossi, M., Cortesogno, L., Galbiati, B., Messiga, B., Piccardo, G. B., & Vanucci, R. (1984). Geologia delle Alpi Liguri: dati, problemi, ipotesi. Memoria della Società Geologica Italiana, 28, 5-57.

Vanossi, M. (1986). Geologia delle Alpi liguri. Mem. Soc. Geol. It., 28, 598.

- Vérati, C., Lardeaux, J. M., Favier, A., Corsini, M., Philippon, M., & Legendre, L. (2018). Arc-related metamorphism in the Guadeloupe archipelago (Lesser Antilles active Island arc): First report and consequences. *Lithos*, 320-321, 592–598.
- Vidal, O., & Parra, T. (2000). Exhumation paths of high-pressure metapelites obtained from local equilibria for chlorite-phengite assemblage. *Geological Journal*, 35, 139–161.
- Vidal, O., De Andrade, V., Lewin, E., Munoz, M., Parra, T., & Pascarelli, S. (2006). P-T deformation Fe2+/Fe3+ mapping at the thin section scale and comparison with XANES mapping: Application to a garnet-bearing metapelite from the Sambagawa metamorphic belt (Japan). Journal of Metamorphic Geology, 24, 669–683.

Warr, L. N. (2021). IMA-CNMNC approved mineral symbols. *Mineralogical Magazine*, 85, 291–320.

Table S1. Representative chlorite EPMA analysis.**Table S2.** Representative phengite EPMA analysis.

SUPPORTING INFORMATION

Additional supporting information may be found in the online version of the article at the publisher's website.

Material S1. Thermobarometry.

Figure S1. Fe in chlorite in the micromap of sample ED1 (intensity map: Fe K α)

Figure S2. K in phengite in the micromap of sample TSE236 (intensity map: K K α)

Figure S3. Distribution of temperature based on Cathelineau (1988) geothermometer in sample TSE236.

How to cite this article: Sanità, E., Di Rosa, M., Lardeaux, J. M., Marroni, M. & Pandolfi, L. (2022). Metamorphic peak estimates of the Marguareis Unit (Briançonnais Domain): New constrains for the tectonic evolution of the southwestern Alps. *Terra Nova*, 00, 1–9. <u>https://doi.org/10.1111/</u> ter.12592